Author’s response to reviews

Title: Intervention Now to Eliminate Repeat Unintended Pregnancy in Teenagers (INTERUPT): a systematic review of intervention effectiveness and cost-effectiveness, and qualitative and realist synthesis of implementation factors and user engagement

Authors:

Rabeea’h Aslam (rwaslam@liverpool.ac.uk)
Maggie Hendry (m.hendry@bangor.ac.uk)
Andrew Booth (a.booth@sheffield.ac.uk)
Ben Carter (CarterBR@cardiff.ac.uk)
Joanna Charles (j.charles@bangor.ac.uk)
Noel Craine (noel.craine@wales.nhs.uk)
Rhiannon Edwards (r.t.edwards@bangor.ac.uk)
Jane Noyes (jane.noyes@bangor.ac.uk)
Lupetu Ntambwe (intambwe@gmail.com)
Diana Pasterfield (d.pasterfield@bangor.ac.uk)
Jo Rycroft-Malone (j.rycroft-malone@bangor.ac.uk)
Nefyn Williams (nefyn.williams@bangor.ac.uk)
Rhiannon Whitaker (Rhiannon@whitres.co.uk)

Version: 1 Date: 07 Feb 2017

Author’s response to reviews:

Dear Mr. Recchioni,

I have updated the following sections as requested.

1. statement of Competing interests
2. copy without comments/track changes, and removed footnotes
3. updated declarations section

If there is anything else, please do let me know.

Thank you for your time

Kind regards,

Rabeea'h